Author's response to reviews

Title: Assessing nonresponse bias at follow-up in a large prospective cohort of relatively young and mobile military service members

Authors:

Alyson J Littman (alyson@u.washington.edu)
Edward J Boyko (eboyko@u.washington.edu)
Isabel G Jacobson (Isabel.Jacobson@med.navy.mil)
Jaime Horton (jaime.horton@med.navy.mil)
Gary D Gackstetter (Gary.Gackstetter@anser.org)
Besa Smith (Besa.Smith@med.navy.mil)
Tomoko Hooper (thooper@usuhs.mil)
Timothy S Wells (tsw_spg@yahoo.com)
Paul J Amoroso (paul.amoroso@us.army.mil)
Tyler C Smith (Tyler.Smith2@med.navy.mil)
Millennium Cohort Study (MilCohortInfo@Med.Navy.Mil)

Version: 8 Date: 22 September 2010

Author's response to reviews:

The requested change to the references has been made.